Author's response to reviews

Title: The spinal posture of computing adolescents in a real-life setting

Authors:

Yolandi Brink (ybrink@sun.ac.za)
Quinette Louw (qalouw@sun.ac.za)
Karen Grimmer (Karem.Grimmer@unisa.edu.au)
Esme Jordaan (Esme.Jordaan@mrc.ac.za)

Version: 3 Date: 6 June 2014

Author's response to reviews:

Editorial comments:
1. Please kindly include the reference number for the ethics approval where appropriate.
I have included the Ethics approval nr in line 86.

2. Figure 3 contains an image of a student. Please confirm if appropriate written informed consent was obtained for publication of their image from the participant or, where participants are children, a parent or guardian. Please ensure that this is reflected in the manuscript.
We realized that we only have permission to publish the photograph in a thesis and therefore we replaced the photograph (figure 3) with a schematic representation of the marker placement and changed the title of the figure in line 514.